Author's response to reviews

Title: A precautionary public health protection strategy for the possible risk of childhood leukaemia from exposure to power frequency magnetic fields

Authors:

Myron Maslanyj (myron.maslanyj@hpa.org.uk)
Tracy Lightfoot (tracy.lightfoot@egu.york.ac.uk)
Joachim Schüz (schuzj@iarc.fr)
Zenon Sienkiewicz (zenon.sienkiewicz@hpa.org.uk)
Alastair McKinlay (alastairmckinlay@ntlworld.com)

Version: 4 Date: 13 October 2010

Author's response to reviews: see over